Reviewer #1:

**Scientific Quality:** Grade A (Excellent)

**Language Quality:** Grade B (Minor language polishing)

**Conclusion:** Accept (High priority)

**Specific Comments to Authors:**
I find this short review excellent. It is informative and helpful to those working with MG patients. Information on specific immune targets for autoantibodies other than ACh receptors is often absent from texts targeting nonspecialists, and is here correctly addressed. Although I have no criticism as to its scientific contents, I suggest the text might become even better if a thorough correction of minor issues in English were carried out.

Author’s Response:

The authors are extremely thankful to the reviewer for sparing the time and effort to review our manuscript and appreciating the content. As suggested language editing has been done by Enago, one of the suggested companies. Certificate is attached. Their modified manuscript is also attached for review.

**Language evaluation:** The English-language grammatical presentation needs to be improved to a certain extent. Before final acceptance, the authors must provide the English Language Certificate issued by a professional English language editing company. Please visit the following website for the professional English language editing companies we recommend: [https://www.wjgnet.com/bpg/gerinfo/240](https://www.wjgnet.com/bpg/gerinfo/240)

Author’s Response:

As suggested language editing has been done by Enago. English language editing has been done by one of the suggested companies. Certificate is attached. Their modified manuscript is also attached for review.